Author's response to reviews

Title: Can standardized patients replace physicians as OSCE examiners?

Authors:

Kevin McLaughlin (kevin.mclaughlin@calgaryhealthregion.ca)
Laura Gregor (lgregor@hotmail.com)
Allan Jones (ajone@ucalgary.ca)
Sylvain Coderre (coderre@ucalgary.ca)

Version: 4 Date: 6 January 2006

Author’s response to reviews: see over
January 6, 2006

BMC Medical Education

Dear Sir/Madam,

Re: Can standardized patients replace physicians as OSCE examiners?

Thank you for accepting this paper pending revisions. I have now made all of the revisions requested, namely:

Title & Abstract - These are identical in the manuscript and in the submission system.

Author list - The list of authors in the manuscript is the same as the submission system.

Title - Sentence case is used in the title.

References - Reference number is now before the full stop.

Authors contributions - Changed to Authors’ contributions.

I hope that this manuscript is now acceptable for publication.

Thanking you in anticipation.

Yours faithfully,

Dr. Kevin McLaughlin
(Corresponding Author)